Author's response to reviews

Title: Assessment of a Continuous Blood Gas Monitoring System in Animals during Circulatory stress.

Authors:

SANDRO GELSOMINO (sandro.gelsomino@libero.it)
ROBERTO LORUSSO (sandro.gelsomino@libero.it)
 STEFANO ROMAGNOLI (sandro.gelsomino@libero.it)
UGOLINO LIVI (sandro.gelsomino@libero.it)
ROCCO CARELLA (sandro.gelsomino@libero.it)
FABIANA LUCA' (sandro.gelsomino@libero.it)
ATTILIO RENZULLI (sandro.gelsomino@libero.it)
GIL BOLOTIN (sandro.gelsomino@libero.it)
GIUSEPPE DE CICCO (sandro.gelsomino@libero.it)
JOS MAESSEN (sandro.gelsomino@libero.it)
PIER LUIGI STEFANO (sandro.gelsomino@libero.it)
GIAN FRANCO GENSINI (sandro.gelsomino@libero.it)

Version: 2 Date: 21 May 2010

Author's response to reviews:

Editor-in-Chief
BMC Biotechnology
Florence, 21/05/2010

Dear Sir,

Please, find herewith enclosed the original clinical paper titled “Assessment of a Continuous Blood Gas Monitoring System in Animals during Circulatory stress” for being considered for publication in BMC Biotechnology.

We declare that: 1) Authors of the study have not conflict of interest 2) All authors have read and approved the manuscript 3) The manuscript is original and it is not under consideration elsewhere.

Looking forward to hearing from you and from the Editorial Board,
Yours Sincerely
On behalf of all the Authors,
Sandro Gelsomino